Author’s response to reviews

Title: Association between plasma adipsin level and mild cognitive impairment in Chinese patients with type 2 diabetes: a cross-sectional study

Authors:

Dan Guo (1596868651@qq.com)
Rong Huang (huangrong0914@yeah.net)
Sai Tian (760128840@qq.com)
Jiaqi Wang (592244572@qq.com)
Hongyan Lin (15850690376@163.com)
Ke An (765611151@qq.com)
Jin Han (13851602376@126.com)
Shaohua Wang (gyjwsh@126.com)

Version: 3 Date: 18 Sep 2019

Author’s response to reviews:

Dear Editors and Reviews,

Firstly, I would like to express my thanks for your help and support regarding revisions to the manuscript entitled "Association between plasma adipsin level and mild cognitive impairment in Chinese patients with type 2 diabetes: a cross-sectional study” (Manuscript Number: BEND-D-19-00187R2), as well as the positive comments from prestigious and respectable reviewers.

We have carefully read the full text again and comments, the precise comments are helpful for us to summarize the current study and promote further research. We have done relevant modifications according to the specific comments suggested by the reviewers. The following pages display the detailed changes and additional remarks about our manuscript.

Finally, again I wish the corrections would be able to match your standard and to publish smoothly. It is our honor to publish in the BMC Endocrine Disorders. Please contact with me without any hesitation if you have further questions.

I am looking forward to your reply.

Best regards,

Prof. Shaohua Wang
We would like to express our sincere thanks to the reviewer for the constructive and positive comments.

Specific Comments and Replies to Reviewers and Editors

Editor Comments:

Before anyone can assess this revision, the authors must submit a manuscript with all changes highlighted in the revision document - with a point-by-response "response to reviewers" document also included.

Thank you very much for your consideration of our manuscript. We have provided a version with all changes highlighted in the revision document and sent each response letter as an additional material. We are not sure if this meets your requirements. If you have any questions, please contact with me without any hesitation.